Author’s response to reviews

Title: Renal limited AL amyloidosis - a diagnostic and management dilemma

Authors:
Kar Wah Fuah (tracymiyako@hotmail.com)
CHRISTOPHER LIM (drchrislim@gmail.com)

Version: 2 Date: 09 Oct 2018

Author’s response to reviews:

Editor Comments:

1. Histology

Can you confirm in the manuscript how many times you performed the different stains on different sections of the tissue? A lot of what you describe may be open to interpretation, and it would be good to know that these observations were repeated in multiple replicate tests.

Answer: All the different stains were repeated twice. We have added this into the manuscript.

2. Authors Contributions

In this section, can you please initialize each author and list their specific contribution to creating and preparing this manuscript. Please can you also make it clear in the manuscript who was responsible for preparing the figure files.

Answer: Thank you and this is indicated now inside the manuscript
Reviewer reports:

Suchin Worawichawong (Reviewer 2):

The additional information could not convince the diagnosis of AL-Amyloid because resistant to KMnO4 treatment Congo red staining is not a diagnostic criteria only for AL-Amyloid but it can be seen in non-AA amyloid.

Answer:

Thank you for your suggestion, we are aware that the limitation of this method is that it cannot exclude familial amyloidosis. However, there is no family history of amyloidosis.